Author's response to reviews

Title: Hepatitis B virus vaccination booster does not provide additional protection in adolescents: A cross-sectional school-based study

Authors:

Yung-Chieh Chang (tonyijane@hotmail.com)
Jen-Hung Wang (paulwang@tzuchi.com.tw)
Yu-Sheng Chen (lister3813401@yahoo.com.tw)
Jun-Song Lin (8887241@ms21.url.com.tw)
Ching-Feng Cheng (chengcf@mail.tcu.edu.tw)
Chia-Hsiang Chu (chuchia@ms6.hinet.net)

Version: 7  Date: 4 September 2014

Author's response to reviews: see over
Dear Editor:

We thank you for the helpful comments. We have revised our manuscript according to your suggestions. We hope this revised manuscript answers all your comments and will be acceptable for publication. Point-by-point replies to the editor’s comments are attached in the following pages.

Best Regards,

Yung-Chieh Chang, M.D.
Fellow of Pediatric Gastroenteroloy
Department of Pediatrics
Hualien Tzu-Chi Medical Center, Taiwan
Authors’ response to editor

Title: Hepatitis B virus vaccination booster does not provide additional protection in adolescents: A cross-sectional school-based study

Authors:
Yung-Chieh Chang (tonyijane@hotmail.com)
Jen-Hung Wang (paulwang@tzuchi.com.tw)
Yu-Sheng Chen (lister3813401@yahoo.com.tw)
Jun-Song Lin (8887241@ms21.url.com.tw)
Ching-Feng Cheng (chengcf@mail.tcu.edu.tw)
Chia-Hsiang Chu (chuchia@ms6.hinet.net)
Editor’s comments:

- Line 46-47 in the Abstract should be changed to the sentence from Line 250 from the main text as it reads clearer on the same point: "Among the 24 recipients showing anti-HBs seroconversion at 6 weeks after booster, seven subjects (29.2\%) had lost their anti-HBs seropositivity again within 3 years."

  Thank you. The sentence from Line 46-47 in Abstract has now been substituted with the sentence from Line 250 from the main text.

- Also delete "Ab" at line 102

  It has been deleted according to your suggestion. Thank you.